Author's response to reviews

Title: Assessment of ePrescription quality: an observational study at three mail-order pharmacies

Authors:

Bengt Åstrand (bengt.astrand@hik.se)
Emelie Montelius (emelie.montelius@hik.se)
Göran Petersson (goran.petersson@hik.se)
Anders Ekedahl (anders.ekedahl@apoteket.se)

Version: 5 Date: 9 January 2009

Author's response to reviews:

To the Editor of BMC Medical Informatics and Decision Making

The manuscript has been checked according to the formatting requirements. Unnecessary capitals have been removed in section headings. The horizontal line has been deleted.

Yours sincerely
Bengt Åstrand
Ph D